Author’s response to reviews

Title: To what degree do patients actively choose their healthcare provider at the point of referral by their GP? A video observation study.

Authors:

Amy Potappel (amypotappel@hotmail.com)
Maartje Meijers (m.meijers@nivel.nl)
Corelien Kloek (corelien.kloek@hu.nl)
Aafke Victoor (a.victoor@nivel.nl)
Janneke Noordman (j.noordman@nivel.nl)
Tim olde Hartman (Tim.OldeHartman@radboudumc.nl)
Sandra van Dulmen (s.vandulmen@nivel.nl)
Judith de Jong (j.dejong@nivel.nl)

Version: 2 Date: 21 Nov 2019

Author’s response to reviews:

To the Editor of BMC Family Practice
Gerard Clunn, PhD

November 21, 2019

Dear Editor, dear dr. Gerard Clunn

Thank you very much for reviewing our manuscript. Below, we provide a point-by-point response to your concerns. We hope that you find our responses satisfactory and that the manuscript is now acceptable for publication in BMC Family Practice. We look forward to your response.

On behalf of all authors,

Yours sincerely,

Aafke Victoor, PhD
Correspondence concerning this manuscript may be addressed to:
Aafke Victoor, PhD
Nivel, Netherlands institute for health services research
P.O. Box 1568
3500 BN Utrecht
The Netherlands
Tel.: +31 30 2729826
E-mail: a.victoor@nivel.nl

1. As a member of the editorial board (Associate Editor) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.

We added “One of the authors (Tim olde Hartman) is Associate Editor at BMC Family Practice” to the Competing Interests section.

2. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

“The funders had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.” is now added to the Funding section.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We uploaded our manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files are also clean versions. Figures (and additional files) are uploaded as separate files.